6.885: Geometric Folding Algorithms Fall 2007 — Prof. Erik Demaine

Problem Set 2

Due: Friday, October 5, 2007

Problem 1. Fold an origami model.

You are free to either come up with your own folding, or fold any origami diagram you find on the web or in books. Submit your model and a short description of the design (title, where it came from, who designed it, etc.).

The purpose of this problem is to gain familiarity with the physicality of paper folding. Your model does not have to be pristine—just do your best.